

G OPEN ACCESS

Citation: Caprioli T, Martindale S, Mengiste A, Assefa D, H/Kiros F, Tamiru M, et al. (2020) Quantifying the socio-economic impact of leg lymphoedema on patient caregivers in a lymphatic filariasis and podoconiosis co-endemic district of Ethiopia. PLoS Negl Trop Dis 14(3): e0008058. https://doi.org/10.1371/journal.pntd.0008058

Editor: Arianna R. Means, University of Washington, UNITED STATES

Received: June 28, 2019

Accepted: January 14, 2020

Published: March 3, 2020

Copyright: © 2020 Caprioli et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevent data by lymphoedema severity group are available in the manuscript and it's supporting information files. Individual level data on socio-economic factors are not freely available for ethical reasons as public availability would comprise patient and caregiver privacy. Data requests may be sent to the Centre for Neglected Tropical Diseases, Liverpool School of Tropical Medicine (CNTD@lstmed.ac.uk) for researchers who meet the criteria for access to confidential data. RESEARCH ARTICLE

Quantifying the socio-economic impact of leg lymphoedema on patient caregivers in a lymphatic filariasis and podoconiosis coendemic district of Ethiopia

Thais Caprioli¹, Sarah Martindale¹, Asrat Mengiste², Dereje Assefa², Fikre H/Kiros², Mossie Tamiru³, Nebiyu Negussu³, Mark Taylor¹, Hannah Betts¹, Louise A. Kelly-Hope¹*

1 Department of Tropical Disease Biology, Centre for Neglected Tropical Diseases, Liverpool School of Tropical Medicine, Liverpool, United Kingdom, 2 National Podoconiosis Action Network, Addis Ababa, Ethiopia, 3 Federal Ministry of Health, Addis Ababa, Ethiopia

* Louise.Kelly-Hope@lstmed.ac.uk

Abstract

Background

Lymphoedema caused by lymphatic filariasis (LF) or podoconiosis can result in physical disability and social exclusion, which is exacerbated by painful acute dermatolymphangioadenitis (ADLA) episodes. These conditions have a significant impact on patients, however, little is known about the indirect effects on their caregivers. This study, therefore, aimed to determine the impact on caregivers for patients with leg lymphoedema in a co-endemic district of Ethiopia.

Methodology/Principal findings

A cross-sectional survey of lymphoedema patients and their caregivers was conducted using semi-structured questionnaires in the Southern Nation Nationalities Peoples Region (SNNPR) of Ethiopia. Lymphoedema patient information on clinical severity (mild, moderate, severe), frequency of ADLAs, their socio-demographic characteristics and the identity of main caregiver(s) was collected. Caregiver information on socio-demographic characteristics, types of care provided, their quality of life (QoL) measured across nine domains, and productivity was collected, with key indicators compared in the presence and absence of patients' ADLAs.

A total of 73 patients and 76 caregivers were included. Patients were grouped by mild/ moderate (n = 42, 57.5%) or severe (n = 31, 42.5%) lymphoedema, and reported an average of 6.1 (Cl \pm 2.18) and 9.8 (Cl \pm 3.17) ADLAs respectively in the last six months. A total of 48 (65.8%) female and 25 (34.2%) male patients were interviewed. Caregivers were predominately male (n = 45, 59.2%), and spouses formed the largest caregiving group for both female and male patients. **Funding:** This research was funded by a grant support from the Department for International Development (DFID) and GlaxoSmithKline (GSK) to the Centre of Neglected Tropical Diseases (CNTD) within the Liverpool School of Tropical Medicine (LSTM). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

In the absence of an ADLA, most caregivers (n = 42, 55.2%) did not provide care, but only one caregiver did not provide care during an ADLA. In the absence of an ADLA, the average time (hour:minute) spent by mild/moderate (00:17, CI: \pm 00:08) and severe (00:10, CI: \pm 00:07) patient caregiver per task was minimal. The time mild/moderate (00:47, CI: \pm 00:11) and severe (00:51, CI: \pm 00:16) patient caregivers spent per task significantly increased in the presence of an ADLA. In addition, caregivers' QoL was negatively impacted when patients experienced an ALDA, and they had to forfeit an average of 6 to 7 work/ school days per month.

Conclusion/Significance

Lymphoedema and ADLAs impact negatively on patients' and their caregivers' lives. This emphasises the importance of increasing access to effective morbidity management and disability prevention services to reduce the burden and help to address the Sustainable Development Goal (SDG) 5, target 5.4, which seeks to recognise and value unpaid care and domestic work.

Author summary

Lymphatic filariasis and podoconiosis can result in chronic severe lymphoedema of the limbs, which is aggravated by frequent painful episodes of acute bacterial infection known as 'acute attacks'. The debilitating impact of these conditions on patients are well documented, however, the extent to which the hardship extends to their caregivers is unknown. To address this knowledge gap, a survey of 73 lymphoedema patients and their caregivers was conducted in an endemic region of Ethiopia. Patients were found to be predominately female and caregivers predominately male, mostly spouses. Caregivers provided a wide range of care, and the type, frequency and time spent on tasks significantly increased when patients experienced an acute attack. Caregivers reported a negative impact on their quality of life and provided care for an average of 15 days and forfeited an average of 7 days of work or school per month. This study has revealed a previously hidden burden of lymphoedema that not only affects patients, but their caregivers who are mostly family members. This emphasises the importance of increasing access to effective morbidity management and disability prevention programmes to reduce the burden and help to address the Sustainable Development Goal (SDG) 5, target 5.4, which seeks to recognise and value unpaid and domestic work.

Introduction

Lymphatic filariasis (LF) and podoconiosis are chronic disabling neglected tropical diseases (NTDs) that affect millions of people worldwide [1]. LF is a mosquito-borne disease caused by the parasites *Wuchereria bancrofti, Brugia malayi* and *B. timori* which are endemic in 73 countries with an estimated 40 million people affected clinically with lymphoedema and hydrocoele [2]. Infections are usually acquired during childhood, but the physical impairments develop later in life due to a dysfunctional and damaged lymphatic system, which causes fluid accumulation in the limbs, breasts and genitalia [3]. Several stages of limb lymphoedema exist, and whilst mild stages are reversible overnight, severe stages are permanent and can result in

significant disability and loss of independence [3–5]. Alongside the disfigurements, patients experience re-current acute dermatolymphangioadenitis (ADLA) known as 'acute attacks', which are episodes of febrile illness, malaise and extensive swelling caused by secondary bacterial infection [6,7]. ADLAs have been identified as a critical factor contributing to the progression of diseases and predictor of disability [8–10]. Productive hours are lost, and the impairments render a socially excluded reality for many, thereby, contributing to the global loss of 2.78 million Disability Adjusted Life Years (DALYS) [11].

Podoconiosis is largely concentrated in Africa, and is known or suspected to be endemic in 32 countries affecting an estimated four million people worldwide [12–14]. Unlike LF, podoconiosis is a non-communicable disease, which is caused by the continuous exposure to irritant soil in barefoot populations [15,16]. Penetrated soil particles initiate the inflammatory cascade, which hinders the lymphatic system and results in the progressive bilateral swelling of the lower limbs. Similar to LF, several stages of lymphoedema exist [4], and the most severe stages are associated with frequent ADLAs and increased disability. The incapacitating and stigmatised impairments fuels severe economic consequences, inter-generational poverty [17] and alienates patients from society, including school, community meetings, churches and mosques [18].

The disability posed by leg lymphoedema, regardless of the aetiology, is likely to extend beyond the patient, to their caregivers. The World Health Organization (WHO) [19] defines a caregiver as: 'A person who provides support. . . with various activities to persons with disabilities or long-term conditions. . . (who) provides emotional or financial support, as well as hands-on help with different tasks'. Caregivers in lower middle-income countries (LMIC) have been characterised as poorly educated middle-aged wives [20]. Although caregiving may be largely rewarding, it can be laden with physical and emotional adversities, collectively known as the 'caregiving burden' [21]. Straining duties and installing the obligation to accomplish all roles, caregiving sustains an incremental build-up of stress [20,22]. Caregiving is time consuming, productive hours are forfeited and caregivers' psychological health is known to suffer [20,23,24]. Furthermore, cognisant of the degree to which the social pathology of the stigmatised impairments afflicts lymphoedema patients [25,26], whether their caregivers suffer additional hardship has yet to be ascertained.

Overall, there is little research on caregivers of people affected by NTDs. It has been estimated that five million LF patients are reliant on caregivers [27]. For limb lymphoedema, the type and frequency of care required is likely to be affected by the severity and the occurrence of ADLAs [28–30]. During ADLAs, due their reduced mobility, patients are likely to rely on greater assistance to perform self-care and day-to-day activities. This was reported in a recent paper which qualitatively outlined the financial and emotional worries instilled in podoconiosis caregivers whilst caring for patients during their ADLAs [31]. Furthermore, the gender of the caregivers can influence the types of care provided [26], women may be more likely to assist in personal care, than men [24]. Understanding the gender roles and it's contributions to the hidden work caregiving embodies is important [32] as it is likely to pose a significant socio-economic burden on families and communities [33]. It is also fundamental to ensuring gender equity, which has been explored in relation to NTD mass drug administration programmes (MDA) [34]. Further, it may help achieve the Sustainable Development Goal (SDG) in relation to NTDs [35], as well as address SDG 5, target 5.4, which seeks to recognise and value unpaid care and domestic work [36].

LF and podoconiosis morbidity management and disability prevention (MMDP) programmes largely consist of limb hygiene and self-care activities, which are paramount to alleviate suffering and to reduce the occurrence of ADLAs [37]. It is pragmatic, cost-effective and efficient to integrate the LF and podoconiosis MMDP programmes where possible [38]. Involving caregivers may optimise the use of human resources of health, encourage the continuity of care and promote societal disability inclusion. In settings where caregivers form the linchpin of informal care, their welfare and, indirectly, their ability to provide care, is pivotal. Hence, a better understanding of who these caregivers are—how they assist- and what impact caregiving has on their quality of life (QoL), needs to be established.

This study, therefore, aimed to determine the socio-economic impact on caregivers for patients with leg lymphoedema in Ethiopia, where an estimated 5.6 million individuals are at risk of LF and 34.9 million individuals are at risk of podoconiosis [38]. A cross-sectional survey in a co-endemic district highlighted caregiver socio-demographic characteristics, types of care provided, QoL, and productivity, with key indicators compared in the presence and absence of patients' ADLAs.

Methods

Ethics statement

Ethical approval was obtained from the Southern Nations Nationalities Peoples Region (SNNPR) Health Bureau and the Research Ethics Committee of the Liverpool School of Tropical Medicine prior the commencement of the study. All survey participants participated on a voluntary basis and informed consent was obtained. Participants who were literate provided written informed consent, and those who were illiterate had the information sheet read to them and oral informed consent was documented. For those aged <18 years, parental or guarantor informed consent was obtained, and they were present during the survey interview.

Study site and characteristics

The study was conducted in the South Ari district of the South Omo Zone, within the SNNPR. The South Ari district comprises of 50 kebeles (villages) and has an estimated population of 265,866 people [39]. In 2015, using a probability-based sampling of school attending children to determine the prevalence of LF antigenemia, the South Ari district was re-mapped for LF. Results showed that 2.1% of immunochromatographic tests (ICTs) were positive, and the district was recommended to undergo MDA [40]. For podoconiosis, it is reported that the SNNPR has a prevalence of 8.63% (Confidence Interval (CI) \pm 0.66%) [41].

Information on the burden of leg lymphoedema was obtained from a community-based integrated morbidity mapping survey conducted in 2015 where the severity of lymphoedema, caused by LF or podoconiosis, was staged as mild, moderate or severe using WHO guidelines [42]. The South Ari district recorded 2,277 lymphoedema patients with mild (n = 1,588, 66.8%), moderate (n = 655, 27.5%) or severe (n = 135, 5.7%) conditions [42]. The Gazer and Metzer health catchment areas were selected for the current study due to the high number of lymphoedema patients. Gazer health catchment recorded a total of 363 patients with mild (n = 250, 68.9%), moderate (n = 94, 25.9%) and severe (n = 19, 5.2%), and Metzer health catchment recorded 273 patients with mild (n = 207, 75.8%), moderate (n = 45, 16.5%), and severe (n = 21, 7.7%) conditions recorded.

Study design, sampling and participants

A cross-sectional survey of lymphoedema patients and their caregivers was conducted over a three-week period in June 2018. Lymphoedema patients were randomly selected from the available database and aimed to include 30 patients with mild, moderate and severe lymphoedema (total n = 90), which would provide a representative sample for each staging group based on an infinite population size, utilising a 90% confidence level and a 15% margin of error. For

bi-lateral lymphoedema patients, the stage of the limb most severely affected was used. The WHO definition of caregiver (19) was used as the study's operational definition of a caregiver. Based on the operational definition, patients identified their main caregiver (s) (maximum of two individuals) and if eligible, were invited to participate in the survey. In both the lymphoedema patients and patient caregivers' questionnaires, an ADLA or an acute attack was defined as 'an episode of acute inflammation, pain, redness and swelling in the limb, associated with lymph nodes and fever'.

Two semi-structured questionnaires, one for the lymphoedema patients and one for the patient caregivers, were administered by the field team which comprised one researcher, two field assistants, and a clinician. The clinician confirmed if the patient's lymphoedema was related to either LF or podoconiosis, assessed the stage of the lymphoedema and elicited information on symptoms related to acute attacks. If the patient or caregiver did not know what an acute attack was, then the clinician would ask if the patient had experienced fever, pain, redness or swelling of their affected limbs. Both questionnaires were translated into Amharic and piloted within the team and conducted in an interview-style. The clinician was fluent in English, Amharic and Arigna (the local dialect), and provided language assistance to participants if required. Data was collected in English using electronic tablets and the Open Data Kit (ODK) software [43].

Lymphoedema patients. Patients with lymphoedema related to LF or podoconiosis, aged >18 years of age, and who were able to identify a caregiver were invited to participate in the survey. The patient questionnaire was administered to obtain information on:

- i. Clinical condition (mild, moderate, severe) including the frequency of ADLA episodes in the past six months (number in total)
- ii. Socio-demographic characteristics (gender, age, marital status, educational attainment)
- iii. Identification of their main caregiver(s) (name, age and relationship to the patient). The information of a maximum of two caregivers per patient was documented.

Patient caregivers. Caregivers who were identified by the patients and aged >10 years were invited to participate in the survey and interviewed separately from their patients. The caregiver questionnaire was administrated to obtain information on:

- i. Socio-demographic characteristics (gender, age, marital status, educational attainment)
- ii. Types of care provided in the absence and presence of an ADLA, which were broadly grouped as 'symptom management' (care to relieve patients' symptoms) and 'day-to-day assistance' (care to mitigate patients' loss of independence). The type of care related to these two groups was further categorised as; one-to-one (requiring direct hands-on assistance), inside the household (indirect care provided inside the household), outside the household (indirect care provided outside the household) and other (summarised in Table 1)
- iii. The time spent providing each reported symptom management and day-to-day assistance task was recorded, both in the absence and presence of an ADLA.
- iv. The QoL questions were adapted from the Lymphatic Filariasis Quality of Life (LFSQQ) questionnaire, which has been validated in previous LF studies [44]. The questionnaire consisted of nine domains, and caregivers ranked them according to a four-point scale (no problem = 1, mild = 2, moderate = 3, severe = 4). The QoL domains included; Planning (had difficulty to plan); Neglect (felt devalued by their community); Health (worried about

	Symptom management tasks	Day-to-day assistance tasks
One-to-one	Washing/drying/massaging/applying creams to the limb (s)	Helping with washing/ dressing / mobility
Inside the household	Boiling water to wash the limb	Cooking, washing, laundry, childcare, cleaning
Outside the household	Seeking health care, purchasing medicine, collecting water to wash the limb(s)	Selling/buying market goods, collecting water/wood, tending animals, farming
Other	Praying, collecting ingredients/providing local remedies, providing money to purchase soap	

1 abic 1. Symptom management care and day-to-day assistance categorise	om management care and day-to-day assistance categoris	rise	see
--	--	------	-----

https://doi.org/10.1371/journal.pntd.0008058.t001

their health); Future (worried about the future); Romantic relationships (difficulties in new/existing romantic relationships); Social problems inside and outside the house (difficulties in participating in social activities inside or outside their house); Household (difficulties completing their household duties) and work/school (difficulties attending their work or school).

- v. The impact on caregivers' productivity (number of days care is provided per month, number of work/school days are forfeited per month)
- vi. Any additional comments provided by the caregivers were summarised as quotes under main themes.

Data analysis

The survey data was exported to and visualised in Microsoft Excel and statistically analysed in SPSS (version 23)[45]. For all statistical tests, a p-value ≤ 0.05 was considered statistically significant.

To compare the socio-demographic information between different lymphoedema severity groups and their caregiver(s), the Chi-squared test was used. The mean and the 95% CI were calculated for the reported frequency of ADLAs in the last six months number of days that care was provided per month and the number of work/school days forfeited per month The symptom management and day-to-day assistance tasks were categorised and compared using the Chi squared test, both in the absence and presence of an ADLA. Further, the types of tasks performed by female and male caregivers were compared using the Chi squared test. The time mild/moderate and severe patient caregivers spent providing each type of care was summed within their respective categories. For each category, the mean time and the 95% CI was calculated and compared in the absence and presence of an ADLA with a Wilcoxon signed-rank test. When comparing the gender of the caregiver and the time spent per task category, a Mann Whitney U test was used.

For the QoL, the overall score was quantified by summing the scores allocated to each domain question, based on the following answer scores; no problem = 1, mild = 2, moderate = 3, severe = 4 [39]. The minimum-maximum overall QoL score per person ranged from 9–36. The QoL mean scores (overall and per domain) of mild/moderate and severe patient caregivers were compared both in the absence and presence of an ADLA with a Mann-Whitney U test.

The additional qualitative comments from caregivers were grouped into themes exploring; the types of care provided (social expectations, demanding aspects of care), impact on their productivity (loss of income, impact on children) and on their QoL (relationship with the patient and family members, social withdrawal, afflicted stigma and psychological stress).

Results

Lymphoedema patients

Clinical conditions. A total of 73 lymphoedema patients participated in the survey and had mild (n = 21, 28.8%), moderate (n = 21, 28.8%) and severe (n = 31, 42.5%) conditions verified. Most patients had bi-lateral lymphoedema (n = 69, 94.5%). Due to limitations in obtaining adequate numbers of lymphoedema patients with mild and moderate conditions as patients had moved away, died or were not available, these two lymphoedema stages were combined to form a mild/moderate group (n = 42, 57.5%), which was compared to the severe group (n = 31, 42.5%) in the analyses. All patients reported experiencing an ADLA in the past six months. On average, mild/moderate patients reported 6.1 (CI \pm 2.18) ADLAs, and severe patients 9.8 (CI \pm 3.17) ADLAs in the last six months.

Socio-demographic characteristics. A total of 48 female (65.8%) and 25 male (34.2%) lymphoedema patients participated in the survey. Patients' mean age was 47.7 years (female 49.2 years (CI \pm 4.88); male 44.8 years (CI \pm 8.80)) and ranged from 20–90 years. Approximately half the patients were married (n = 39, 53%), with one fifth divorced (n = 15, 20.5%) or wid-owed (n = 15, 20.5%). Most patients were illiterate with no school education (n = 54, 74%). Overall, there was no significant difference in socio-demographic characteristics between the mild/moderate and severe patient group, except by gender, where a higher proportion of female patients had mild/moderate condition than severe, and a higher proportion of male patients had severe than mild/moderate (p = 0.29) (S1 Table).

Main caregivers. A total of 76 caregivers were identified, of which 45 (59.2%) provided care to mild/moderate patients and 31 (40.8%) provided care to severe patients. Three patients reported having two caregivers. Spouses formed the largest caregiving group for both female and male patients as shown in Fig 1A and 1B with 40% (n = 20) of female patients being cared



for by their husband, and 54% of male patients by their wife (n = 14). Sons were the second main caregiver group for both female (n = 12, 24%) and male (n = 6, 23%) patients. Other caregivers included daughters, church leaders, neighbours, sons-in-law and nieces.

Patient caregivers

Socio-demographic information. Of the 76 caregivers interviewed, 31 (40.8%) were female and 45 (59.2%) were male. Caregivers' mean age was 37.6 years (female 38 years (CI \pm 5.95); male 37.3 years (CI \pm 4.76)) and ranged from 13–80 years. Three caregivers were children aged 13, 15 and 17 years. The majority of caregivers were married (n = 61, 80.3%) and illiterate with no schooling (n = 48, 63.2%). Overall, there was no significant difference in socio-demographic characteristics between mild/moderate and severe patient caregivers (S2 Table). One severe patient caregiver reported that their patient did not experience ADLAs, and hence was not included in the analysis.

Types of care. The two types of care and tasks that patient caregivers provided in the absence and in the presence of an ADLA are summarised in Table 2 (A maximum of two tasks were recorded per caregiver). Overall, in the absence of an ADLA, 25 (55.6%) mild/moderate and 17 (56.7%) severe patient caregivers provided no help to manage patients' symptoms or to assist their day-to-day activities. However, this reduced significantly in the presence of an ADLA with all, except one caregiver, needing to provide help (Table 2).

Symptom management. For symptom management care, in the absence of an ADLA the majority of mild/moderate (n = 33, 73%) and severe (n = 21, 70%) patient caregivers provided no help. However, this changed significantly in the presence of an ADLA as shown in Table 2. There was a significant increase in patient caregivers providing help one-to one and outside the house for both the mild/moderate and severe patients. This significant change is illustrated in Fig 2A-2D. In the presence of an ADLA, 18 caregivers (mild/moderate n = 12; severe: n = 6) performed multiple tasks. There was no significant difference in the types of care and the tasks provided by the gender of the patient caregiver. However, a higher proportion of female caregivers provided support with one-to-one tasks (mild/moderate: 46.7%; severe: 33.3%) than male caregivers (mild/moderate: 23.3%; severe: 13.3%) (Table 3).

•	••		-									
Types of care	Tasks ^b	Mild/m	fild/moderate (N = 45)Severe (N						$N = 30)^{a}$			
		No ADI	LA	ADLA		p-value	No ADL	A	ADLA		p-value	
		n	%	n	%		n	%	n	%		
Symptom management	No help	33	73.3	4	8.9	0.000*	21	70.0	5	16.7	0.000*	
	One-to-one	5	11.1	14	31.1	0.020*	0	0.0	7	23.3	0.005*	
	Inside	1	2.2	1	2.2	1.000	6	20.0	6	20.0	1.000	
	Outside	5	11.1	32	71.1	0.000*	2	6.7	19	63.3	0.000*	
	Other	2	4.4	6	13.3	0.138	2	6.7	3	10.0	0.640	
Day-to-day assistance	No help	31	68.9	7	15.6	0.000*	24	80.0	7	23.3	0.000*	
	One-to-one	2	4.4	3	6.7	0.645	0	0.0	1	3.3	0.313	
	Inside	6	13.3	18	40.0	0.004*	3	10.0	12	40.0	0.007*	
	Outside	10	22.2	26	57.8	0.001*	5	16.7	20	66.7	0.000*	
Overall no help		25	55.6	1	2.2	0.000*	17	56.7	0	0.00	0.000*	

Table 2. A summary of the types of care and the tasks caregivers provided.

* p≤0.05

^a One caregiver was removed from the analysis as their patient did not suffer from ADLA episodes

^b A maximum of two tasks were recorded per caregiver

https://doi.org/10.1371/journal.pntd.0008058.t002

100

90

80

70

60

50

40

30

20

10

0

Percentage

B. Mild/moderate - presence of ADLA







C. Severe – absence of ADLA

inside

outside

other

one to one



D. Severe - presence of ADLA



Symptom management

Day-to-day activities

Fig 2. Comparison of the types of care provided by the mild/moderate and severe patient caregivers in the absence and presence of an ADLA.

https://doi.org/10.1371/journal.pntd.0008058.g002

Table 3. A summary of the types of care and	the tasks	s provided by	the gender o	of the careg	iver in the	e presence of	an ADLA
---	-----------	---------------	--------------	--------------	-------------	---------------	---------

Types of care	Tasks ^b	Mild/moderate (N = 45)					Severe $(N = 30)^{a}$				
		Female (N = 15)	Male (N = 30)	V	p-value	Female (N = 15)	Male (N = 15)		p-value
		n	%	n	%		n	%	n	%	
Symptom management	One to one	7	46.7	7	23.3	0.111	5	33.3	2	13.3	0.195
	Inside	1	6.7	0	0.00	0.153	5	33.3	1	6.7	0.068
	Outside	10	66.7	22	73.3	0.642	9	60.0	10	66.7	0.705
	Other	0	0.00	6	20.0	0.063	1	6.7	2	13.3	0.543
Day-to-day assistance	One to one	2	13.3	1	3.3	0.205	1	6.7	0	0.00	0.309
	Inside	6	40.0	12	40.0	1.000	7	46.7	5	33.3	0.456
	Outside	5	33.3	21	70.0	0.019	10	66.7	10	66.7	1.000

^a One caregiver was removed from the analysis as their patient did not suffer from ADLA episodes

^b A maximum of two tasks were recorded per caregiver

https://doi.org/10.1371/journal.pntd.0008058.t003

Types of care	Tasks ^b	Mild/mode	5)		Severe $(N = 30)^{a}$						
		No ADLA		ADLA		p-value	No ADLA		ADLA		p-value
		Time (hr:min)	± CI	Time (hr:min)	± CI		Time (hr:min)	± CI	Time (hr:min)	± CI	
Symptom management	One-to-one	00:03	00:03	00:08	00:05	0.038*	00:00	00:00	00:09	00:07	0.016*
	Inside	00:01	00:01	00:01	00:01	1.000	00:04	00:03	00:04	00:03	1.000
	Outside	00:11	00:12	01:46	00:40	0.000^{*}	00:04	00:06	01:31	00:40	0.000*
	Other	00:06	00:11	00:12	00:13	0.059	00:01	00:01	00:03	00:05	0.180
	Total	00:05	00:04	00:32	00:12	0.000^{*}	00:02	00:02	00:27	00:12	0.000*
Day-to-day assistance	One-to-one	00:01	00:02	00:04	00:07	0.285*	00:00	00:00	00:02	00:04	0.317
	Inside	00:27	00:23	01:08	00:34	0.014*	00:21	00:24	01:06	00:42	0.073
	Outside	01:13	00:49	02:06	00:50	0.012*	00:39	00:44	03:02	01:22	0.009*
	Total	00:34	01:13	01:06	01:01	0.000^{*}	00:20	00:59	01:23	01:16	0.001*
Overall mean time spent		00:17	00:08	00:47	00:11	0.000*	00:10	00:07	00:51	00:16	0.000*

Table 4. A summary of t	he mean time spent per categor	y (hr: min) in both types of ca	are, both in the absence and	presence of an ADLA
	1 1 0			1

 $* p \leq 0.05$

^a One caregiver was removed from the analysis as their patient did not suffer from ADLA episodes

^b The time of a maximum of two tasks were recorded per caregiver

https://doi.org/10.1371/journal.pntd.0008058.t004

Day-to-day assistance. For day-to-day assistance, in the absence of an ADLA, the majority of mild/moderate (68.8%) and severe (80%) patient caregivers provided no help. However, this changed significantly in the presence of an ADLA as shown in Table 2. There was a significant increase in patient caregivers providing help inside and outside the house for both the mild/moderate and severe patients. This significant change is illustrated in Fig 2A–2D. In the presence of an ADLA, 42 caregivers (mild/moderate n = 27; severe: n = 15) performed multiple tasks. There was no significant difference in the types of care and the tasks provided by the gender of the patient caregiver. However, a higher proportion of female caregivers provided support with one-to-one tasks than male caregivers (Table 3).

Caregiving time. Overall, in the absence of an ADLA, mild/moderate (mean: 00:17 (hours (hr):minutes (min)), CI = $\pm 00:08$ (hr:min) and severe (mean: 00:10 (hr:min), CI = $\pm 00:07$ (hr:min)) patient caregivers spent little time in providing help to manage patient's symptoms and assist their day-to-day activities. However, this significantly increased in the presence of an ADLA (mild/moderate mean: 00:47 (hr:min), CI = $\pm 00:11$ (hr:min); severe mean: 00:51 (hr:min), CI = $\pm 00:16$ (hr:min)). During an ADLA, all patient caregivers increased the amount of time per task to help manage patients' care.

Symptom management. For symptom management care, in the absence of an ADLA, patient caregivers spent the longest time providing help for tasks outside the household mild/ moderate: mean = 00:11 (hr:min), CI: \pm 00:12 (hr:min); severe: mean = 00:04 (hr:min), CI: \pm 00:06 (hr:min)), which increased significantly in the presence of an ADLA (mild/moderate: mean = 01:46 (hr:min), CI: \pm 00:12 (hr:min); severe: mean = 01:31 (hr:min), CI: \pm 00:40 (hr: min)) (Table 4). There was no significant difference in the types of care and the tasks provided by the gender of the patient caregiver. However, a higher proportion of female caregivers spent more time providing care for one-to-one tasks than male caregivers (Table 5).

Day-to-day assistance. For day-to-day assistance, in the absence of an ADLA, patient caregivers spent the longest time providing help for tasks outside the household (mild/moderate: mean = 00:13 (hr:min), CI: $\pm 00:49$ (hr:min); severe: mean = 00:39 (hr:min), CI: $\pm 00:44$ (hr:min)), which increased significantly in the presence of an ADLA (mild/moderate:

Types of care	Tasks ^b	Mild/moderate (N	Severe $(N = 30)^{a}$								
		Female (N = 15)		Male $(N = 3)$	Male (N = 30) p-1		alue Female (N = 15)		Male (N = 15)		p-value
		Time (hr:min)	± CI	Time (hr:min)	± CI		Time (hr:min)	± CI	Time (hr:min)	± CI	
Symptom management	One to one	00:13	00:12	00:06	00:05	0.112	00:11	00:10	00:06	00:09	0.250
	Inside	00:03	00:06	00:00	00:00	0.157	00:07	00:06	00:01	00:03	0.080
	Outside	01:26	01:12	01:58	00:54	0.509	01:15	00:50	01:48	01:07	0.536
	Other	00:00	00:00	00:18	00:19	0.067	00:04	00:09	00:03	00:04	0.605
	Total	00:24	00:16	00:35	00:16	0.897	00:24	00:14	00:29	00:19	0.476
Day-to-day assistance	One to one	00:13	00:28	00:00	00:01	0.191	00:04	00:09	00:00	00:00	0.317
	Inside	00:58	01:10	01:01	00:41	0.663	01:02	00:43	01:09	01:19	0.574
	Outside	00:50	01:18	02:30	01:06	0.055	02:35	01:53	03:29	02:12	0.641
	Total	00:57	00:33	01:10	00:28	0.399	01:14	00:42	01:33	00:54	0.772
Overall mean time spent		00:39	00:17	00:50	00:15	0.470	00:45	00:2 <mark>0</mark>	00:57	00:25	0.440

Table 5. A summary of the mean time each spent per task category (hr: min) in both types of care during an ADLA.

*p≤0.05

^a One caregiver was removed from the analysis as their patient did not suffer from ADLA episodes

^b The time of a maximum of two tasks were recorded per caregiver

https://doi.org/10.1371/journal.pntd.0008058.t005

mean = 02:35 (hr:min), CI: \pm 01:53 (hr:min); severe: mean = 03:29 (hr:min), CI: \pm 02:12 (hr: min)) (Table 4). There was no significant difference in the types of care and the tasks provided by the gender of the patient caregiver. However, a higher proportion of female caregivers spent more time providing care for one-to-one tasks than male caregivers (Table 5).

Quality of Life

Overall in the absence of an ADLA, the overall mean QoL score (mild/moderate: score = 1.4; severe: score = 1.4) and the mean QoL score for each domain (mild/moderate range = 1.1-1.7,

Domains	Mild/moderate (N = 4	5)		Severe $(N = 30)^{a}$		
	QoL sco	ores ^b		QoL sco	ores ^b	
	No ADLA	ADLA	p-value	No ADLA	ADLA	p-value
Plans	1.4	3.0	0.000*	1.2	3.1	0.000*
Neglect	1.7	1.9	0.387	1.5	2.0	0.260
Future	1.6	2.5	0.005*	1.5	2.5	0.000*
Health	1.5	2.4	0.017*	1.5	2.5	0.110
Romantic relationships	1.1	1.9	0.003*	1.3	1.7	0.363
Social outside	1.1	3.1	0.000*	1.1	2.8	0.000*
Social inside	1.2	3.2	0.000*	1.2	3.0	0.000*
Household	1.4	3.5	0.000*	1.5	3.6	0.000*
Work/school	1.4	3.6	0.000*	1.4	3.6	0.000*
Mean of overall score	1.4	2.8	0.000*	1.4	2.8	0.000*

Table 6. A summary of caregivers' mean score per QoL domain.

*p≤0.05

^a One caregiver was removed from the analysis as their patient did not suffer from ADLA episodes

^b Category defined by the range of the mean QoL Score per domain: No problem: 1–1.4; Mild: 1.5–2.5; Moderate: 2.5–3.4; Severe: 3.5–4

https://doi.org/10.1371/journal.pntd.0008058.t006

severe range = 1.2-1.5) were low, reflecting no or mild problems in caregiving (Table 6). However, this changed in the presence of an ADLA, and the overall mean QoL score (mild/moderate: score = 2.8; severe: score = 2.8) and the mean QoL score for each domain (mild/moderate range = 1.9-3.6, severe range = 1.7-3.6) increased significantly, reflecting moderate to severe problems in caregiving. For each of the domains, the mean QoL score was low in the absence of an ADLA and significantly increased in the presence of an ADLA. The domains; supporting the household and attending work/school, observed the greatest difference in the mean QoL score between the absence and the presence of an ADLA.

The changes in QoL across the nine domains are illustrated in Fig 3A and 3B for mild/moderate patient caregivers and Fig C-D for severe patient caregivers. The proportion of caregivers reporting a severe problem with their QoL ranged from 1–7% in the absence of an ADLA (mild/moderate range 1–3%; severe 1–7%) which then increased significantly to 8–34% (mild/ moderate 8–25%; severe 10–34%).

Impact on productivity

Per month, mild/moderate and severe patient caregivers provided care on average for 14.8 (CI \pm 3.31) and 15.5 (CI \pm 4.33) days respectively. No significant difference was found between the severity groups (p = 0.813). Per month, caregivers forfeited a similar amount of work/school days (mild/moderate: 7.17; (CI \pm 1.92); severe 5.87; (CI \pm 1.89)).





D. Severe - presence of ADLA



Additional comments

There were three main themes related to the additional comments that caregivers provided and included i) types of care ii) impact on productivity and iii) QoL

Types of care: <u>Social expectations</u>: Two male caregivers reported that socially constructed gender norms acted as a barrier when providing certain types of care.

"The culture prevents me to help my wife..." (Male, 60)

'Many do not understand the caring duties I have' (Male, 28)

<u>Demands of caregiving</u>: Many caregivers (n = 14) reported challenges related to their poverty.

'I want to take her to the clinic, but I am too poor' (Male, 70)

'Her situation and my poverty stress me out. I do not know how to help' (Male, 29)

Impact on productivity. Loss of income: Many caregivers reported financial problems

(n = 35) and decreased food security (n = 7).

'We do not have enough money for food' (Male, 28)

'During her acute attack she stays in bed for 7 days, so I stay with her and lose 7 days of farming. Our family unit is becoming very poor' (Male, 29)

<u>Impact on children</u>: Many caregivers (n = 11) reported not being able to send their children to school.

'The children also suffer... unable to go to school as we have no money for the uniforms' (Male, 60)

'When I see my friends, they are all in eighth grade. I worry about my future' (Female, 15) Quality of life. <u>Relationship with the patient</u>: Many found caregiving rewarding (n = 26),

however, many also reported their relationship with the patient to be deteriorating, notably during ADLAs (n = 14).

'We exchange many blessings' (Female, 65)

'I feel like I have lost my husband' (Female, 60)

<u>Relationship with other family members:</u> Two caregivers reported that their relationship with other family members had been impacted.

'I have separated from my husband to provide care to my mother' (Female, 35)

'I feel like I have lost my 'normal wife' (Male, 36)

<u>Social withdrawal</u>: Many caregivers (n = 29) reported that they had no time to socialise with friends or to contribute to community activities.

'Many (friends) do not understand the caring duties I have' (Male, 28)

'As an Ethiopian we are encouraged to participate in many social activities, the community life is very important to me' (Female, 37)

<u>Afflicted stigma</u>: Two caregivers reported being victim of afflicted stigma or discrimination.

`…my mother's condition is contagious, and people avoid me.' (Male, 29)

"... why God has placed such an illness on our family" (Male, 19)

<u>Psychological stress</u>: Many (n = 9) suffered seeing their relative suffer, and two caregivers felt unable to cope with the patients' suicidal thoughts during ADLAs.

'During an acute attack my wife gets suicidal thoughts. I am scared to leave her alone' (Male, 28)

... if I do not provide enough care, she threatens she will commit suicide' (Male, 65)

Discussion

Lymphoedema patients were found to be predominately female, middle aged and illiterate, which aligns with LF studies elsewhere [28,29], and podoconiosis studies conducted in similar Ethiopian settings [16,46]. Patients predominately had bi-lateral limb lymphoedema, which is

largely associated with podoconiosis related lymphoedema [12]. This present study did not differentiate between the aetiology of the lymphoedema as it sought to quantify the impact of providing care to patients suffering from both LF and podoconiosis related lymphoedema. However, differential diagnosis may be important in certain circumstances, and previous studies have utilised tests such as filarial antigen testing, filarial antibody and parasitological examination to exclude an LF diagnosis in LF and podoconiosis co-endemic areas [15].

All patients experienced frequent ADLAs, with a higher frequency of ADLAs in patients with severe stage lymphoedema [6,7,42], which contribute to the progression of disease and are key predictors of disability [8–10]. Whilst recall bias may have influenced the number of ADLAs reported, patients with mild/moderate and severe lymphoedema experienced an average of 6 and 10 respectively, which may be broadly extrapolated to 12 to 20 ADLAs per year. This aligns with the findings of a recent paper, which documented that podoconiosis patients with persistent lymphoedema experienced an average of 22 ADLA episodes a year [47]. Regardless of the lymphoedema severity, whilst experiencing an ADLA, patients' increased reliance on informal care, from their caregivers, to manage their symptoms and day-to-day activities was observed.

Caregivers were found to be predominately male, middle aged, illiterate and a family member of the patient. The caregivers' age and illiteracy in this study aligns with previous LMIC caregiving research, but wives and daughters were more often identified as LMIC caregivers [20,48]. However, several caregiving studies have identified that patients' spouses often became their caregivers [23], which may explain this study's higher proportion of male caregivers and lack of daughter caregivers providing care to male patients. Although this study's small sample size limits the generalisability, the gender and role of caregivers is likely to be determined by the patient's disease, gender and care needs, which may vary greatly by the clinical nature of the disease, cultural and social factors, and epidemiological distribution. This is currently not well understood in the context of NTDs and more research on different diseases in different settings are required to obtain a more comprehensive picture and to monitor equity [33].

Caring for patients during ADLAs was found to have a significant impact on caregivers, who needed to provide more support for symptom management and day-to-day tasks, including help with one-to-one tasks, such as limb washing. This is important as the WHO recommends regular limb hygiene to mitigate the morbidity associated with lymphoedema and enabling the continuity of such tasks during an ADLA, demonstrates caregivers' indispensable role as a human resource in providing the MMDP programmes' minimum package of care [10,12,47]. In addition, tasks outside the household, which often mitigated the household's economic loss, such as farming or tending the animals, were prominently provided by caregivers to assist patients' day-to-day activities during an ADLA.

Our study found some differences by gender in the types of day-to-day care tasks provided. Tasks inside the household, such as cooking, and one-to-one care, were more often performed by female caregivers. Studies in comparable settings have documented similar caregiving tasks and related to traditional feminine role [49,50]. Tasks outside the household were more often performed by male caregivers. Understanding the different gender roles in caregiving may help to achieve the SDG 5, target 5.4, which seeks to value informal domestic work [36]. Albeit, it was challenging to distinguish between activities which females were 'expected' to perform on a routine basis [26] and those which were executed in their capacity as a caregiver.

Patients' productivity during ADLAs and its contribution to their household's economic downfall has been documented [17,51]. However, our research helps to uncover and quantify a hidden aspect to the economic burden that family units suffer and suggests that previous evaluations have grossly underestimated the magnitude of the burden that lymphoedema poses on households' economy. The hardship experienced by family units is not only due to

the patients decreased working hours, but, in union with caregivers' lack of productivity whilst providing care. Similarly, a study exploring the impact of ADLAs on podoconiosis patient caregivers in Ethiopia, reported that, at times, caregivers forfeited work to care for their relative [31]. Although there may be recall bias in remembering the frequency and impact of ADLAs, studies in Tanzania [52] and Malawi [29] have reported that patients lose between three and eleven days per ADLA episode, which aligns with our study where caregivers reported losing an average of six to seven work or school days per month. Forfeiting productive hours, often minimises time to farm and curtails household profit that can strain food security [32].

Caregiving inflicts a multitude of stressors, which collectively contribute to a decreased QoL [24]. We found that when patients experienced an ADLA, the caregivers' QoL was negatively impacted. Notably, caregivers found more difficulties with participating in social activities inside or outside their house, completing their household duties and attending their work or school. The impact on socialising may be due to time constraints or socially constructed barriers, such as stigma linked to the diseases [26] or to the gender-bias largely associated with caregiving [49]. The impact on child caregivers not attending school is of a particular concern with long-term implications, which urgently need to be addressed.

This study reveals the concerning state of LF and podoconiosis patient caregivers, a previously hidden dimension to the public health burden posed by NTDs, which identifies the need of accessible support services. Caregivers' role in achieving the global aims of eliminating and mitigating the disability caused by these diseases needs to be better understood and supported. The study highlights the need to scale-up effective MMDP programmes to reduce the overall burden of disease and occurrence of ADLAs, thereby, alleviating the suffering of patients and reducing the impact on their caregivers, which will also help to address the SDG targets.

Supporting information

S1 STROBE Checklist. (DOC)

S1 Table. A comparison of mild/moderate and severe patients' socio-demographic information.

(DOCX)

S2 Table. A comparison of mild/moderate and severe patient caregivers' socio-demographic information. (DOCX)

Acknowledgments

The authors are grateful for all the participants who took part in the study and to the in-field team and in-country collaborator in Ethiopia. The authors would also like to express thanks to the Health Extension Workers, who facilitated data collection.

Author Contributions

Conceptualization: Louise A. Kelly-Hope.

Data curation: Thais Caprioli.

Formal analysis: Thais Caprioli, Sarah Martindale, Hannah Betts.

Funding acquisition: Mark Taylor.

Investigation: Thais Caprioli, Sarah Martindale, Fikre H/Kiros, Mossie Tamiru, Nebiyu Negussu.

Methodology: Sarah Martindale, Asrat Mengiste, Hannah Betts, Louise A. Kelly-Hope.

Project administration: Thais Caprioli, Sarah Martindale, Dereje Assefa, Fikre H/Kiros.

Supervision: Sarah Martindale, Asrat Mengiste, Hannah Betts, Louise A. Kelly-Hope.

Visualization: Thais Caprioli, Hannah Betts, Louise A. Kelly-Hope.

Writing – original draft: Thais Caprioli, Louise A. Kelly-Hope.

Writing – review & editing: Thais Caprioli, Sarah Martindale, Asrat Mengiste, Dereje Assefa, Fikre H/Kiros, Mossie Tamiru, Nebiyu Negussu, Mark Taylor, Hannah Betts, Louise A. Kelly-Hope.

References

- 1. Organisation WH. Integrating neglected tropical diseases into global health and development [Internet]. Geneva: World Health Organisation; 2017. Available: https://apps.who.int/iris/bitstream/handle/10665/ 255011/9789241565448-eng.pdf
- 2. World Health Organization. Lymphatic Filariasis [Internet]. 2019 [cited 4 Jul 2019]. Available: <u>https://www.who.int/lymphatic_filariasis/en/</u>
- Dreyer G, Norões J, Figueredo-Silva J, Piessens WF. Pathogenesis of lymphatic disease in bancroftian filariasis: A Clinical Perspective. Parasitol Today. 2000; 16: 544–548. https://doi.org/10.1016/s0169-4758(00)01778-6 PMID: 11121854
- Douglass J, Kelly-Hope L. Comparison of Staging Systems to Assess Lymphedema Caused by Cancer Therapies, Lymphatic Filariasis, and Podoconiosis. Lymphat Res Biol. 2019; <u>https://doi.org/10.1089/lrb.</u> 2018.0063 PMID: 30789319
- Zeldenryk LM, Gray M, Speare R, Gordon S, Melrose W. The emerging story of disability associated with lymphatic filariasis: A critical review. PLoS Negl Trop Dis. 2011; 5(12): e1366. https://doi.org/10. 1371/journal.pntd.0001366 PMID: 22216361
- Pani SP, Yuvaraj J, Vanamail P, Dhanda V, Michael E, Grenfell BT, et al. Episodic adenolymphangitis and lymphoedema in patients with bancroftian filariasis. Trans R Soc Trop Med Hyg. 1995; 89: 72–74. https://doi.org/10.1016/0035-9203(95)90666-5 PMID: 7747314
- 7. Pani SP, Srividya A. Clinical manifestations of bancroftian filariasis with special reference to lymphoedema grading. Indian J Med Res. 1995; 102: 114–118. PMID: 8543349
- Shenoy RK, Sandhya K, Suma TK, Kumaraswami V. A preliminary study of filariasis related acute adenolymphangitis with special reference to precipitating factors and treatment modalities. Southeast Asian J Trop Med Public Health. 1995; 26: 301–305.
- Dreyer G, Addiss D, Bettinger J, Dreyer P, Noroes J, Rio F. Lymphoedema Staff Manual—Treatment and Prevention of Problems Associated with Lymphatic Filariasis [Internet]. World Health Organisation (WHO); 2001. Available: http://apps.who.int/iris/bitstream/handle/10665/67224/WHO_CDS_CPE_ CEE_2001.26a.pdf?sequence=1
- Stocks ME, Freeman MC, Addiss DG. The Effect of Hygiene-Based Lymphedema Management in Lymphatic Filariasis-Endemic Areas: A Systematic Review and Meta-analysis. PLoS Negl Trop Dis. 2015; 9: 1–19. https://doi.org/10.1371/journal.pntd.0004171 PMID: 26496129
- Murray CJL, Vos T, Lozano R, Naghavi M, Flaxman AD, Michaud C, et al. Disability-adjusted life years (DALYs) for 291 diseases and injuries in 21 regions, 1990–2010: a systematic analysis for the Global Burden of Disease Study 2010. Lancet. 2012; 380: 2197–2223. https://doi.org/10.1016/S0140-6736 (12)61689-4 PMID: 23245608
- 12. World Health Organization (WHO). Podoconiosis: endemic non-filarial elephantiasis [Internet]. 2019. Available: https://www.who.int/lymphatic_filariasis/epidemiology/podoconiosis/en/ Acessed 15/11/2019
- 13. Deribe K, Cano J, Trueba ML, Newport MJ, Davey G. Global epidemiology of podoconiosis: A systematic review. PLoS Negl Trop Dis. 2018; 12: e0006324–e0006324. <u>https://doi.org/10.1371/journal.pntd.</u> 0006324 PMID: 29494642
- Tekola Ayele F, Adeyemo A, Finan C, Hailu E, Sinnott P, Burlinson ND, et al. HLA class II locus and susceptibility to podoconiosis. N Engl J Med. 2012; 366: 1200–1208. https://doi.org/10.1056/ NEJMoa1108448 PMID: 22455414

- Deribe K, Beng AA, Cano J, Njouendo AJ, Fru-Cho J, Awah AR, et al. Mapping the geographical distribution of podoconiosis in Cameroon using parasitological, serological, and clinical evidence to exclude other causes of lymphedema. PLoS Negl Trop Dis. 2018; 12: e0006126. https://doi.org/10.1371/journal.pntd.0006126 PMID: 29324858
- Deribe K, Brooker SJ, Pullan RL, Sime H, Gebretsadik A, Assefa A, et al. Epidemiology and individual, household and geographical risk factors of podoconiosis in Ethiopia: results from the first nationwide mapping. Am J Trop Med Hyg. 2015; 92: 148–158. https://doi.org/10.4269/ajtmh.14-0446 PMID: 25404069
- Yordanos M, Tomczyk S, Amberbir T, Tamiru A, Davey G. Patients' perceptions of podoconiosis causes, prevention and consequences in East and West Gojam, Northern Ethiopia. BMC Public Health. 2012; 828. https://doi.org/10.1186/1471-2458-12-828 PMID: 23020758
- Tora A, Davey G, Tadele G. A qualitative study on stigma and coping strategies of patients with podoconiosis in Wolaita zone, Southern Ethiopia. Int Health. 2011; 3: 176–181. https://doi.org/10.1016/j.inhe. 2011.06.006 PMID: 24038367
- World Health Organization. A glossary of the terms for community health care and services for older persons [Internet]. WHO/WKC/Tech.Ser./04.2 Collections; 2004. Available: <u>http://www.who.int/kobe_centre/ageing/ahp_vol5_glossary.pdf</u>
- 20. Thrush A, Hyder A. The neglected burden of caregiving in low- and middle-income countries. Disabil Heal. 2014; 7: 262–272. https://doi.org/10.1016/j.dhjo.2014.01.003 PMID: 24947567
- **21.** Braithwaite V. Caregiving burden: Making the concept scientifically useful and policy relevant. Res Aging. 1992; 14: 3–27.
- Shibre T, Kebede D, Alem A, Negash A, Deyassa N, Fekadu A, et al. Schizophrenia: illness impact on family members in a traditional society-rural Ethiopia. Soc Psychiatry Psychiatr Epidemiol. 2003; 38: 27–34. https://doi.org/10.1007/s00127-003-0594-7 PMID: 12563556
- 23. Lambert S, Bowe S, Livingston P, Heckel L, Cook S, Kowal P, et al. Impact of informal caregiving on older adults' physical and mental health in low-income and middle-income countries: a cross-sectional, secondary analysis based on the WHO's Study on global AGEing and adult health (SAGE). BMJ Open. 2017; 7: 1–14.
- Pinquart M, Sörensen S. Gender differences in caregiver stressors, social resources, and health: An updated meta-analysis. J Gerontol B Psychol Sci Soc Sci. 2006; 61: 33–45.
- Asampong E, Dako-Gyeke M, Oduro R. Caregivers' views on stigmatization and discrimination of people affected by leprosy in Ghana. PLoS Negl Trop Dis. D; 2018; 12: e0006219. https://doi.org/10.1371/ journal.pntd.0006219 PMID: 29377890
- Dako-Gyeke M. Courtesy stigma: A concealed consternation among caregivers of people affected by leprosy. Soc Sci Med. 2018; 196: 190–196. <u>https://doi.org/10.1016/j.socscimed.2017.11.030</u> PMID: 29195190
- 27. Ton TG, Mackenzie C, Molyneux D. The burden of mental health in lymphatic filariasis. Infect Dis Poverty. BioMed Central; 2015; 4: 34. https://doi.org/10.1186/s40249-015-0068-7
- Budge PJ, Little KM, Mues KE, Kennedy ED, Prakash A, Rout J, et al. Impact of community-based lymphedema management on perceived disability among patients with lymphatic filariasis in Orissa State, India. PLoS Negl Trop Dis. 2013; 7: e2100. https://doi.org/10.1371/journal.pntd.0002100 PMID: 23516648
- Martindale S, Mkwanda SZ, Smith E, Molyneux D, Stanton MC, Kelly-Hope LA. Quantifying the physical and socio-economic burden of filarial lymphoedema in Chikwawa District, Malawi. Trans R Soc Trop Med Hyg. 2014; 108. https://doi.org/10.1093/trstmh/tru154 PMID: 25270880
- Stanton MC, Yamauchi M, Mkwanda SZ, Ndhlovu P, Matipula DE, Mackenzie C, et al. Measuring the physical and economic impact of filarial lymphoedema in Chikwawa district, Malawi: a case-control study. Infect Dis Poverty. London: BioMed Central; 2017; 6: 28. <u>https://doi.org/10.1186/s40249-017-0241-2</u> PMID: 28366168
- Phillips C, Samuel A, Tiruneh G, Deribe K, Davey G. The impact of acute adenolymphangitis in podoconiosis on caregivers: A case study in Wayu Tuka woreda, Oromia, Western Ethiopia. "If she was healthy, I would be free." PLoS Negl Trop Dis. United States; 2019; 13: e0007487. https://doi.org/10. 1371/journal.pntd.0007487 PMID: 31283763
- Samman E, Prester-Marshall E, Jones N, Bhatkal T, Melamed C, Stavropoulou M, et al. Women's work; Mothers, children and the global childcare crisis [Internet]. London.: Overseas Development Institute; 2016. Available: https://www.odi.org/publications/10349-women-s-work-mothers-childrenand-global-childcare-crisis
- **33.** Engels D, Savioli L. Reconsidering the underestimated burden caused by neglected tropical diseases. Trends Parasitol. 2006; 22: 363–366. https://doi.org/10.1016/j.pt.2006.06.004 PMID: 16798088

- Theobald S, MacPherson E, Dean L, Jacobson J, Ducker C, Gyapong M, et al. 20 years of gender mainstreaming in health: lessons and reflections for the neglected tropical diseases community. BMJ Glob Heal. 2017; 2(4): e000512. https://doi.org/10.1136/bmjgh-2017-000512 PMID: 29177100
- Fitzpatrick C, Engels D. Leaving no one behind: A neglected tropical disease indicator and tracers for the Sustainable Development Goals. Int Health. 2015; 8: i15–i18. https://doi.org/10.1093/inthealth/ ihw002 PMID: 26940304
- 36. United Nations. Global indicator framework for the Sustainable Development Goals and targets of the 2030 Agenda for Sustainable Development [Internet]. 2015. Available: https://unstats.un.org/sdgs/indicators/GlobalIndicatorFrameworkafterrefinement_Eng.pdf
- World Health Organization. Morbidity management and disability prevention in lymphatic filariasis [Internet]. Geneva; 2013. Available: https://www.who.int/lymphatic_filariasis/resources/9789241505291/en/
- Deribe K, Kebede B, Tamiru M, Mengistu B, Kebede F, Martindale S, et al. Integrated morbidity management for lymphatic filariasis and podoconiosis, Ethiopia. Bull World Heal Organ. 2017; 95: 652–656.
- Central Statistical Authority. 2007 Population and houseing census of Ethiopia [Internet]. Addis Ababa, Ethiopia.; 2012. Available: http://unstats.un.org/unsd/censuskb20/Attachment489.aspx? AttachmentType=1 Acessed 15/11/2019
- 40. Gass KM, Sime H, Mwingira UJ, Nshala A, Chikawe M, Pelletreau S, et al. The rationale and cost-effectiveness of a confirmatory mapping tool for lymphatic filariasis: Examples from Ethiopia and Tanzania. PLoS Negl Trop Dis. United States; 2017; 11: e0005944. https://doi.org/10.1371/journal.pntd.0005944 PMID: 28976981
- Deribe K, Cano J, Newport MJ, Golding N, Pullan RL, Sime H, et al. Mapping and Modelling the Geographical Distribution and Environmental Limits of Podoconiosis in Ethiopia. Public Library of Science; 2015;
- 42. Kebede B, Martindale S, Mengistu B, Kebede B, Mengiste A, Kiros FH, et al. Integrated morbidity mapping of lymphatic filariasis and podoconiosis cases in 20 co- endemic districts of Ethiopia. PLoS Negl Trop Dis. 2018; 12(7): e0006491. https://doi.org/10.1371/journal.pntd.0006491 PMID: 29965963
- Brunette Sudar S., Sundt M., Larson C., Beorse J., Anderson R. W. Open Data Kit 2.0: A Services-Based Application Framework for Disconnected Data Management [Internet]. Mobisystems. 2017. doi: http://dx.doi.org/10.1145/3081333.3081365
- Cristina T, Saravu RN, Kuthaje SB, Kuthaje V, Steven N, Dennis PW, et al. Comparison of three quality of life instruments in lymphatic filariasis: DLQI, WHODAS 2.0, and LFSQQ. PLoS Negl Trop Dis. 2014; e2716. https://doi.org/10.1371/journal.pntd.0002716 PMID: 24587467
- 45. IBM Corp. Released 2016. IBM SPSS Statistics for Windows NY: IBM Corp. V 24. 0. A. IBM SPSS for window, version 24.0.
- 46. Molla YB, Le Blond JS, Wardrop N, Baxter P, Atkinson PM, Newport MJ, et al. Individual correlates of podoconiosis in areas of varying endemicity: a case-control study. PLoS Negl Trop Dis. 2013; 7: e2554. https://doi.org/10.1371/journal.pntd.0002554 PMID: 24340109
- Negussie H, Molla M, Ngari M, Berkley J, Kivaya E, Njuguna P, et al. Lymphoedema management to prevent acute dermatolymphangioadenitis in podoconiosis in northern Ethiopia (GoLBeT): a pragmatic randomised controlled trial. Lancet Glob Heal. 2018; e795. <u>https://doi.org/10.1016/S2214-109X(18)</u> 30124-4 PMID: 29773516
- **48.** Martindale S, Mackenzie C, Mkwanda S, Smith E, Stanton M, Molyneux D, et al. " Unseen " Caregivers: The disproportionate gender balance and role of females in the home-based care of lymphatic filariasis patients in Malawi. Front Women's Heal. 2017; 2: 1–3. https://doi.org/10.15761/FWH.1000126
- **49.** Sharma N, Chakrabarti S, Grover S. Gender differences in caregiving among family- caregivers of people with mental illnesses. World J Psychiatry. 2016; 6: 7–17. <u>https://doi.org/10.5498/wjp.v6.i1.7</u> PMID: 27014594
- Robinson CA, Bottorff JL, Pesut B, Oliffe JL, Tomlinson J. The Male Face of Caregiving: A Scoping Review of Men Caring for a Person With Dementia. Am J Mens Heal. 2014/01/09. 2014; 8: 409–426. https://doi.org/10.1177/1557988313519671 PMID: 24414033
- Ramaiah KD, Das PK, Michael E, Guyatt HL. The economic burden of lymphatic filariasis in India. Parasitol Today. 2000; 16: 251–253. https://doi.org/10.1016/s0169-4758(00)01643-4 PMID: 10827432
- Gasarasi DB, Premji ZG, Mujinja PGM, Mpembeni R. Acute adenolymphangitis due to bancroftian filariasis in Rufiji district, south east Tanzania. Acta Trop. 2000; 75: 19–28. <u>https://doi.org/10.1016/s0001-706x(99)00090-x PMID: 10708003</u>